

APPENDIX 4.2 -HABITATS REGULATIONS ASSESSMENT

> HAMBLE AIRFIELD HAMBLE-LE-RICE HAMPSHIRE

NOVEMBER 2021, UPDATED OCTOBER 2022

ON BEHALF OF CEMEX



The Old Squash Court, Rempstone Hall Rempstone, Corfe Castle, Wareham, Dorset, BH20 5JQ

#### Tel: 01929 477115 E-mail: enquiries@ecological-services.co.uk

Authorisation

	Name	Date
Report prepared by	JP	10.11.2021
Report checked and	ARH	11.11.2021
authorised by:		
Report amended and updated	JP	23.11.2021
by		
Report checked and	ARH	29.11.2021
authorised by:		
Report amended and updated	CMP & ARH	05.10.2022
by		
Report checked and	LC	06.10.2022
authorised by:		

The contents of this report were correct at the time of the last survey visit. The report is provided for the sole use of the named client and is confidential. The Plans provided within the report are intended to convey information relating to ecology (records, features, impacts, mitigation) and reliance should not be placed on them in terms of the detail of the proposed development.

All rights in this report are reserved. No part of it may be reproduced or transmitted, in any form or by any means, electronic, mechanical, photocopying, recording or otherwise, or stored in any retrieval system of any nature, without our written permission. Its content and format are for the exclusive use of the addressee in dealing with this. It may not be sold, lent, hired out or divulged to any third party not directly involved in this situation without our written consent. It is company policy to share species records collected during our surveys with local biological records centres unless instructed otherwise by the client.

## CONTENTS

1.0	INTRODUCTION	1
2.0	LEGISLATIVE CONTEXT AND THE TESTS OF THE HABITATS	
	REGULATIONS	2
3.0	DEVELOPMENT PROPOSALS AND SCOPE OF ASSESSMENT	4
4.0	BASELINE	6
5.0	LIKELY SIGNIFICANT EFFECT TEST	11
6.0	APPROPRIATE ASSESSMENT	28
7.0	CONCLUSIONS	30
8.0	REFERENCES	32
APPE	NDIX I: APPLICATION SITE LOCATION AND BOUNDARIES	34
APPE	NDIX II: METHOD OF WORKING PLANS (OVERVIEW)	35
APPE	NDIX III: SITE LOCATION IN RELATION TO STATUTORY SITES	36

## 1.0 Introduction

- 1.1 LC Ecological Services Limited were commissioned by CEMEX UK to conduct a Habitats Regulations Assessment for land at the former Hamble Airfield, Hamble Lane, Hamble-le-Rice, Eastleigh, Hampshire, and to devise a mitigation strategy. This was required to support a planning application for phased aggregate extraction on the site, including the erection of a processing plant together with silt lagoons and associated infrastructure, as well as post-quarrying restoration of the land. The application site location and boundaries are depicted on the plan included as appendix I. The Method of Working (MOW) plans are included as appendix II.
- 1.2 The site location lies within 320 metres of three statutory designated terrestrial sites within the national site network (NSN). The three terrestrial sites are: Solent and Southampton Water Special Protection Area (SPA), Solent and Southampton Water Ramsar, and Solent Maritime Special Area of Conservation (SAC). The River Hamble, which lies approximately 410 metres to the east of the site, also forms part of the Solent and Dorset Coast SPA, a maritime SPA designated to protect the foraging habitat of breeding terns. The location of the site relative to these sites is shown on the plan provided in appendix III.
- 1.3 The use of a 2 kilometre area of search to identify sites within the NSN which may be impacted by the proposed project was considered suitable due to the nature of the proposals. Given the distance of the proposed works from the nearest NSN site, direct impacts such as noise and visual disturbance, dust generation and water pollution could potentially result in likely significant effects on NSN sites. Indirect impacts such as recreational impacts or emissions from traffic on NSN sites are considered to be less likely.
- 1.4 These sites receive statutory protection under the Conservation of Habitats and Species Regulations 2017 (as amended) (the 'Habitats Regulations'). The Habitats Regulations afford a high level of protection to sites classified as areas that hold significant populations of certain bird species (SPAs). They also afford the same level of high protection to tracts of land supporting habitats or rare species (other than birds) considered scarce or vulnerable at a European community level (SACs).
- 1.5 Ramsar sites are designated as wetlands of international importance that are afforded similar legislative protection to SPAs and SACs. The government has issued policy statements relating to the special status of Ramsar sites. This extends the same protection afforded to SPAs and SACs.
- 1.6 Under the Habitats Regulations, Hampshire County Council (HCC) is a competent authority, responsible for ensuring that development management decisions do not adversely affect the integrity of NSN sites. This document provides information for the Habitats Regulations Assessment that HCC will need to undertake in determining the planning application for the site. This document considers the implications of the project for the conservation objectives of the four NSN sites to determine whether the project will have an adverse effect on the integrity of the sites, either alone, or in combination with other plans and projects.

# 2.0 Legislative context and the tests of the Habitats Regulations

- 2.1 SACs and SPAs form part of a network of nature protection areas within the UK known as the National Site Network (NSN) and are protected in the determination of a planning application. Under Regulation 63 of the Habitats Regulations the competent authority is responsible for assessing whether land use plans or proposed developments could adversely affect a NSN site. This requires a process known as a Habitats Regulations Assessment (HRA), encompassing two tests required under Regulation 63(1) of the Habitats Regulations.
  - **Test 1:** having ascertained that the plan is not directly connected to, or necessary for site management for nature conservation, the first test of the HRA, commonly referred to as a screening test, considers whether or not a plan or project is likely to have a significant effect on an NSN site either alone or in combination with other plans or projects. A significant effect is any effect that would undermine the conservation objectives for the respective NSN site and may include physical loss and/or damage of a habitat, disturbance effects, and changes to water availability, deposition of contaminants through changes in air quality etc.
  - **Test 2:** The second test of the HRA is relevant to those plans or projects that are screened as likely to have a significant effect alone or in combination with other plans or projects, and requires an appropriate assessment. The role of the appropriate assessment is to consider the implications of the plan or project for the conservation objectives of the NSN sites in question, and to determine whether it will have an adverse effect on the integrity of the site. In carrying out an appropriate assessment, a local authority must have regard to the manner in which the project is proposed to be carried out, or to any conditions or restrictions subject to which it proposes that the consent, permission or other authorisation should be given.
- 2.2 A likely significant effect (LSE) is any effect that is likely to undermine the site's conservation objectives, in light of the characteristics and specific environmental conditions of the SAC/SPA. The likely significant effect test must be based on objective information and the risks must be real, not hypothetical (Boggis vs Natural England 2009).
- 2.3 A recent European Court Judgment (ECJ) People Over Wind and Sweetman v Coillte Teoranta (C-323/17) has altered the process of screening for likely significant effects by overturning the 2008 Hart District Council vs. Secretary of State judgment (2008), known as Dilley Lane. The Dilley Lane judgment stated "there is no legal requirement that a screening assessment... must be carried out in the absence of any mitigation measures that form part of that plan or project".
- 2.4 The recent People Over Wind and Sweetman ruling states that "*it is not appropriate, at the screening stage, to take account of measures intended to avoid or reduce the harmful effects of the plan or project on that site*". This means that mitigation measures must be excluded from the assessment of whether a project is likely to have a significant effect, either alone or in combination with other plans and projects.
- 2.5 In line with the ECJ ruling in Briels (2014) the adverse effect on integrity test that forms part of the appropriate assessment can *"take account of the protective measures forming*

part of the project aimed at avoiding or reducing any direct adverse effects for the site in order to avoid any adverse effects on integrity".

- 2.6 The ECJ ruling in Grace and Sweetman (C-164/17) highlights that a measure can only be considered mitigation "where it is certain it will make an effective contribution to avoiding harm, guaranteeing beyond all reasonable doubt no adverse effect".
- 2.7 Guidance produced by the UK government highlights key principles to be considered by the competent authority when considering if an appropriate assessment is required. It notes that measures that have been specifically added to achieve the purpose of reducing its harmful effects on a habitats site should not be considered at the screening stage.
- 2.8 The government guidance notes that "the scope and content of an appropriate assessment will depend on the nature, location, duration and scale of the proposed plan or project and the interest features of the relevant site. 'Appropriate' is not a technical term. It indicates that an assessment needs to be proportionate and sufficient to support the task of the competent authority in determining whether the plan or project will adversely affect the integrity of the site".

#### Conservation objectives

- 2.9 Conservation objectives are identified for all NSN sites and cover all features that qualify the site for classification or designation. The conservation objectives apply under the Habitats Regulations, Habitats Directive and Wild Birds Directive, and must be considered during a Habitats Regulations Assessment, including an Appropriate Assessment.
- 2.10 For Ramsar sites, a decision has been made by Defra and Natural England not to produce Conservation Advice packages, focusing instead on the production of High Level Conservation Objectives because it's considered that conservation advice available for overlapping European Management Sites is sufficient to support the management of Ramsar interests<sup>1</sup>. European Marine Sites (EMSs) are those areas below mean high water designated as Special Areas of Conservation (SACs) and Special Protection Areas (SPAs).
- 2.11 The Solent European Marine Sites (SEMS) is one of a number of European marine sites in the UK that are designated as internationally important sites for their habitats and species. SEMS covers the harbours, estuaries, areas of open coast and inshore water around the Solent and includes the SSW SPA and the SM SAC. As a matter of policy, the provisions of the Habitats Regulations relating to Habitat Regulations Assessments (HRAs) extend to Ramsar sites. For the purposes of this assessment, the conservation objectives of the SSW SPA and SM SAC are considered to address all relevant interest features in the corresponding Ramsar site.

<sup>&</sup>lt;sup>1</sup> <u>https://www.gov.uk/government/publications/conservation-advice-for-marine-protected-areas-project-background/marine-conservation-advice-project-summary</u>

## 3.0 Development proposals and scope of assessment

- 3.1 The site is located to the north of the village of Hamble-le-Rice and it has an area of approximately 62 hectares. The existing residential areas of Hamble-le-Rice lie to the south and south-east of the site, the B3397 Hamble Lane forms the north-western boundary, Satchell Lane marks the north-eastern boundary, and the site is bounded to the north by the Portsmouth to Southampton railway line.
- 3.2 The site is identified in the adopted Hampshire County Council Minerals and Waste Plan (2013) for extraction of 1.5 million tonnes of sharp sand and gravel.
- 3.3 The development will consist of the phased extraction of sand and gravel working from the north to the south of the site along the western edge and then north again along the eastern edge of the site. As the minerals are extracted from the site inert restoration material will then be imported and used to backfill extraction voids and restore previous ground levels. The existing topsoil layers within each working phase of the project will also be used to restore the site, having first been either used temporarily to form the perimeter bunding, or temporarily stored on site. The nature of the imported restoration material is likely to comprise soils, clay and rubble from construction and demolition sites where the material cannot be recycled. The importation of the inert restoration material will require an environmental permit from the Environment Agency.
- 3.4 It is proposed that the development would commence in 2023 and take up to 13 years to complete. Over the period the site is worked a phased restoration strategy will be implemented with quarried areas backfilled with inert restoration material and capped with the re-used topsoils from the site as the works progress. The restoration of the site will include the re-establishment of similar vegetation cover to what is currently present on site, as well as some new habitats and features.
- 3.5 The proposals include the construction of an earth bank (bunding) around the perimeter of the site to screen the works and provide acoustics mitigation. There will also be a stand-off buffer zone of between 14 and 103 metres width from the outside edge of the earth bunding to the site perimeter, within which existing habitats will be retained and enhanced and new habitats created as part of the ecological mitigation strategy. In addition to this, there will also be a substantial permissive footpath running alongside the northern and eastern boundaries of the site (provided within the stand-off buffer zone). Although these measures have not been specifically included to reduce impacts on the interest features of the SSW SPA and SM SAC they are considered to and therefore require the undertaking of an appropriate assessment.
- 3.6 The site infrastructure would include a site office and processing plant for the washing and grading of gravel, weighbridge, wheel wash area and car-parking. The site is expected to employ approximately 7 full-time staff.
- 3.7 HGV traffic movements would peak in years 3 to 7 when inert material is being imported into the site and extracted minerals exported. In years 8 to 13 HGV traffic movements would reduce as only importation of inert material to the site would

occur during this period. HGV movements in years 1 and 2 would be related only to export of minerals from the site.

- 3.8 Section 4 of this document outlines the interest features of the four sites within the NSN that fall within 2km of the application boundary. Section 5 sets out the likely significant effects assessment where objective information is used to determine if the proposed development will, in the absence of mitigation measures, result in any effect that is likely to undermine the designated site's conservation objectives, in light of the characteristics and specific environmental conditions of the SAC/SPA/Ramsar site.
- 3.9 Section 5 also sets out the rationale for the exclusion of certain likely significant effects from further assessment at this stage. The likely significant effect test must deal with risks that are real, not hypothetical. Potential plausible risk pathways are examined, but if no risk of likely significant effects is identified, these risks are discounted from further assessment.
- 3.10 Section 6 is the appropriate assessment of the likely significant effects of the scheme on interest features of the relevant NSN sites as identified in Section 5. Section 6 deals with impacts from the proposals alone and in-combination with other plans and projects.
- 3.11 Section 7 sets out the conclusions of the shadow appropriate assessment provided by the applicant for the benefit of the competent authority. In undertaking its own Habitats Regulations Assessment of the proposed project, the competent authority will form its own view on the impacts of the scheme on the NSN.

## 4.0 Baseline

4.1 The following section sets out the location, designation criteria and conservation objectives of the NSN sites to be included in this HRA. The locations of these sites relative to the application site are shown in appendix III. This section also provides a brief summary of the bird populations recorded within the site (during baseline field survey work) which are relevant to this assessment.

#### Solent and Southampton Water SPA/Ramsar (SSW SPA)

4.2 The SSW SPA extends from Hurst Spit to Hill Head along the south coast of Hampshire, and from Yarmouth to Whitecliff Bay along the north coast of the Isle of Wight. The site comprises a series of estuaries and harbours with extensive mud-flats and saltmarshes together with adjacent coastal habitats including saline lagoons, shingle beaches, reedbeds, damp woodland and grazing marsh. The mud-flats support beds of *Enteromorpha* spp. and *Zostera* spp. and have a rich invertebrate fauna that forms a food resource for the estuarine birds. In summer, the site is of importance for breeding seabirds, including gulls and four species of terns. In winter, the SPA holds a large and diverse assemblage of waterbirds, including geese, ducks and waders. Dark-bellied brent geese (*Branta bernicla bernicla*) also feed in surrounding areas of agricultural land outside the SPA. The proposals lie within 0.89 kilometres of the SSW SPA.

#### Qualifying features

- 4.3 The Solent and Southampton Water SPA qualifies under Article 4.1 of the Directive (79/409/EEC) as it is used regularly by 1% or more of the GB population of a species listed on Annex I in any season<sup>2</sup>:
  - Mediterranean gull (*Larus melanocephalus*) 2 pairs representing 8.2 13.9% of the GB breeding population (5 year peak mean. Count years 1994-1998).
  - Little tern (*Sterna albifrons*) 49 pairs representing 2% of the GB breeding population (5 year peak mean. Count years 1993-1997).
  - Roseate tern (*Sterna dougallii*) 2 pairs representing 3.1% of the GB breeding population (5 year peak mean. Count years 1993-1997).
  - Common tern (*Sterna hirundo*) 267 pairs representing 2.2% of the GB breeding population (5 year peak mean. Count years 1993-1997).
  - Sandwich tern (*Sterna sandvicensis*) 231 pairs representing 1.7% of the GB breeding population (5 year peak mean. Count years 1993-1997).
- 4.4 The breeding bird assemblage of the SSW SPA is largely confined to saltmarsh habitats west of the Beaulieu River or nature reserves such as Titchfield Haven. The Mediterranean gull colony in Langstone Harbour is currently the largest colony in Hampshire that appears to attract most of the breeding birds from along the Solent coast. The breeding tern colonies are mostly associated with salt-marsh habitats or artificial lagoons and are largely inaccessible during the breeding

<sup>&</sup>lt;sup>2</sup> <u>https://designatedsites.naturalengland.org.uk/Marine/MarineSiteDetail.aspx?SiteCode=UK9011061</u>

season.

- 4.5 The site also qualifies under Article 4.2 of the Directive (79/409/EEC) because it used regularly by 1% or more of the biogeographic population of a regularly occurring migratory species (other than those listed on Annex I) in any season:
  - Eurasian teal (*Anas crecca*) (North-western Europe) 4400 representing 1.1% of the population (5-year peak mean, 1992/3-1996/7).
  - Dark-bellied brent goose (*Branta bernicla bernicla*) (Western Siberia/Western Europe) 7506 representing 2.5% of the population (5-year peak mean, 1992/3-1996/7).
  - Ringed plover (*Charadrius hiaticula*) (Europe/Northern Africa wintering) 552 representing 1.1% of the population (5-year peak mean, 1992/3-1996/7).
  - Black-tailed godwit (*Limosa limosa islandica*) (Iceland breeding) 1125 representing 1.6% of the population (5-year peak mean, 1992/3-1996/7).
- 4.6 The site is also regularly used by over 20,000 waterfowl (as defined by the Ramsar Convention) in any season. Five-year peak mean 51361 (1992/93 1996/97).
- 4.7 Solent and Southampton Water also qualifies as a Ramsar site under four criteria (1, 2, 5 and 6):
  - **Criterion 1**: The site is one of the few major sheltered channels between a substantial island and mainland in European waters, exhibiting an unusual strong double tidal flow and has long periods of slack water at high and low tide. It includes many wetland habitats characteristic of the biogeographic region: saline lagoons, saltmarshes, estuaries, intertidal flats, shallow coastal waters, grazing marshes, reedbeds, coastal woodland and rocky boulder reefs.
  - Criterion 2: The site supports an important assemblage of rare plants and invertebrates. At least thirty three (33) British Red Data Book invertebrates and at least eight (8) British Red Data Book plants are represented.
  - Criterion 5: The site contains avian assemblages of international importance whereby species with peak counts in winter are 51,343 waterfowl (5-year peak mean 1998/99-2002/2003).
  - **Criterion 6**: A wetland should be considered internationally important if it regularly supports 1% of the individuals in a population of one species or subspecies of waterbird.

Species with peak counts in winter are:

- Black-tailed godwit (Iceland/W Europe) 1,240 individuals, representing an average of 2.6% of the population (5-year peak mean 1998/9-2002/3);
- Dark-bellied brent goose- 6,456 individuals, representing an average of 3.2% of the population (5-year peak mean 1998/9-2002/3); and
- Eurasian teal (NW Europe) 5,514 individuals, representing an average of 1.1% of the population (5-year peak mean 1998/9-2002/3).

Species with peak counts in spring/autumn are:

 Ringed plover (Europe/Northwest Africa) – 397 individuals, representing an average of 1.2% of the GB population (5-year peak mean 1998/9-2002/3).

#### Conservation objectives

4.8 The conservation objectives of the SSW SPA are set out in a Natural England publication<sup>3</sup> and are set for each qualifying feature for which the site is classified. Where the objectives are met, the site will be considered to exhibit a high degree of integrity and to be contributing to achieving the aims of the Wild Birds Directive.

With regard to the SPA and the individual species and/or assemblage of species for which the site has been classified (the 'Qualifying Features' listed), and subject to natural change:

Ensure that the integrity of the site is maintained or restored as appropriate, and ensure that the site contributes to achieving the aims of the Wild Birds Directive, by maintaining or restoring:

- The extent and distribution of the habitats of the qualifying features,
- The structure and function of the habitats of the qualifying features,
- The supporting processes on which the habitats of the qualifying features rely,
- The population of each of the qualifying features, and,
- The distribution of the qualifying features within the site.

#### Solent Maritime Special Area of Conservation (SM SAC)

4.9 The proposals lie over 1.2 kilometres south-east from the closest point of the SM SAC. The Annex I habitats that are a primary reason for selection of this site are:

#### 1130 Estuaries

The Solent encompasses a major estuarine system on the south coast of England with four coastal plain estuaries (Yar, Medina, King's Quay Shore, Hamble) and four bar-built estuaries (Newtown Harbour, Beaulieu, Langstone Harbour, Chichester Harbour). The site is the only one in the series to contain more than one physiographic sub-type of estuary and is the only cluster site. The Solent and its inlets are unique in Britain and Europe for their hydrographic regime of four tides each day, and for the complexity of the marine and estuarine habitats present within the area. Sediment habitats within the estuaries include extensive estuarine flats, often with intertidal areas supporting eelgrass (*Zostera* spp.) and green algae, sand and shingle spits, and natural shoreline transitions. The mudflats range from low and variable salinity in the upper reaches of the estuaries to very sheltered almost fully marine muds in Chichester and Langstone Harbours. Unusual features include the presence of very rare sponges in the Yar estuary and a sandy 'reef' of the polychaete (*Sabellaria spinulosa*) on the steep eastern side of the entrance to Chichester Harbour.

<sup>&</sup>lt;sup>3</sup> Natural England, 2014: European Site Conservation Objectives for Solent & Southampton Water Special Protection Area. Site Code: UK9011061.

#### 1320 Spartina swards (Spartinion maritimae)

The Solent contains the second-largest aggregation of Atlantic salt meadows in south and south-west England. Solent Maritime is a composite site composed of a large number of separate areas of saltmarsh. In contrast to the Severn estuary, the salt meadows at this site are notable as being representative of the ungrazed type and support a different range of communities dominated by sea-purslane (*Atriplex portulacoides*), common sea-lavender (*Limonium vulgare*) and thrift (*Armeria maritima*). As a whole, the site is less truncated by man-made features than other parts of the south coast and shows rare and unusual transitions to freshwater reedswamp and alluvial woodland as well as coastal grassland. Typical Atlantic salt meadow is still widespread in this site, despite a long history of colonisation by cord-grass (*Spartina* spp).

- 4.10 Annex I habitats present as a qualifying feature, but not a primary reason for selection of this site:
  - o 1110 Sandbanks which are slightly covered by sea water all the time
  - o 1140 Mudflats and sandflats not covered by seawater at low tide
  - o 1150 Coastal lagoons (priority feature)
  - 1210 Annual vegetation of drift lines
  - 1220 Perennial vegetation of stony banks
  - 1310 Salicornia and other annuals colonising mud and sand
  - 2120 "Shifting dunes along the shoreline with Ammophila arenaria ("white dunes")
- 4.11 The Annex II species 1016 Desmoulin's whorl snail (*Vertigo moulinsiana*) is present as a qualifying feature, but not a primary reason for site selection.

#### Conservation objectives

- 4.12 The conservation objectives of the SM SAC are set out in a Natural England publication<sup>4</sup> and are set for each qualifying feature Annex 1 habitat & Annex 2 species for which the site is classified. Where the objectives are met, the site will be considered to exhibit a high degree of integrity and to be contributing to achieving Favourable Conservation Status for that habitat type at a UK level. The term 'favourable conservation status' is defined in Article 1 of the Habitats Directive. With regard to the SAC and the natural habitats and/or species for which the site has been designated (the 'Qualifying Features' listed below), and subject to natural change; Ensure that the integrity of the site is maintained or restored as appropriate, and ensure that the site contributes to achieving the Favorable Conservation Status of its Qualifying Features, by maintaining or restoring:
  - The extent and distribution of qualifying natural habitats and habitats of qualifying species,

<sup>&</sup>lt;sup>4</sup> Natural England, 2014: European Site Conservation Objectives for Solent Maritime Special Area of Conservation. Site Code: UK0030059.

- The structure and function (including typical species) of qualifying natural habitats,
- The structure and function of the habitats of qualifying species,
- The supporting processes on which qualifying natural habitats and the habitats of qualifying species rely,
- The populations of qualifying species, and
- The distribution of qualifying species within the site.

#### Solent & Dorset Coast Special Protection Area (SDC SPA)

- 4.13 The Solent and Dorset Coast Special Protection Area (SPA) was classified in January 2020 to protect important foraging areas at sea used by terns from colonies within adjacent, already classified, SPAs. The qualifying interest features of the SPA are common tern, Sandwich tern and little tern. Solent and Dorset Coast SPA qualifies under Stage 1.1 by regularly supporting more than 1% of the GB population of Sandwich tern, common tern and little tern, species listed in Annex I of the Birds Directive<sup>5</sup>.
- 4.14 The SPA covers all areas to the mean high-water mark in Portsmouth Harbour, sub-tidal areas with Southampton Water and the River Hamble (below the mean low water mark) and to the mean high-water mark along the coast where terns are not already a qualifying feature of existing SPAs between Worbarrow Bay in Dorset and Bognor Regis in West Sussex. It does not cover the sub-tidal areas of Langstone and Chichester Harbour where the landward boundary is formed by the mean low water as breeding terns are already a feature of the Chichester and Langstone Harbours SPA.
- 4.15 Conservation objectives for the SPA were published in February 2020 by Natural England following the classification of the site. With regard to the potential SPA and the individual species and/or assemblage of species for which the site may be classified and subject to natural change these are to ensure that the integrity of the site is maintained or restored as appropriate, and ensure that the site contributes to achieving the aims of the Wild Birds Directive, by maintaining or restoring:
  - The extent and distribution of the habitats of the qualifying features,
  - The structure and function of the habitats of the qualifying features,
  - The supporting processes on which the habitats of the qualifying features rely,
  - The population of each of the qualifying features, and,
  - The distribution of the qualifying features within the site.
- 4.16 The application site does not support any habitat suitable to support foraging terns. The birds associated with the SPA would be feeding or roosting along the River Hamble.

<sup>&</sup>lt;sup>5</sup> Natural England (2016) Solent and Dorset Coast potential Special Protection Area (pSPA). Departmental brief.

Results of wintering bird surveys.

- 4.17 The former Hamble airfield was identified as having the potential to support wintering brent geese and waders from the Solent in the 2010 Solent Wader and Brent Goose Strategy. At the time the site was classified as uncertain for negative use by waders and there were no records of brent geese. The current version of the strategy has removed the site from the strategy. Data provided by the Hampshire Biodiversity Information Centre (HBIC) in 2021 confirms there are only negative records (confirmed absence of waders and/or brent geese) for this site during surveys.
- 4.18 Wintering bird surveys of the site have been undertaken during the winters of 2015/2016 and 2017/2018. A further update survey, which commenced in October 2021, will conclude in March 2022. The only SPA/Ramsar species recorded during these surveys are a single little egret (*Egretta garzetta*) flying over the site, a flock of four black-tailed godwit flying over, a single common snipe (*Gallinago gallinago*) flushed from the site during one survey visit and a record of two great crested grebe (*Podiceps cristatus*) flying over the site. No brent geese have been recorded on site during any of the wintering bird surveys.
- 4.19 The findings of the wintering bird surveys are in line with the current Solent Waders and Brent Goose Strategy 2020 which does not identify this area as being used by SPA/Ramsar species. On the basis of the field survey results to date and the present condition of the habitats on site, it is concluded that the land is not functionally linked to the SPA/Ramsar.

## 5.0 Likely significant effect test

- 5.1 The first test of Regulation 63 of the Habitats Regulations requires an assessment of whether there are elements of the scheme that are likely to have a significant effect on the NSN sites in question, either alone or in combination with other plans and projects.
- 5.2 As set out in paragraph 2.2 a likely significant effect (LSE) is any effect that is likely to undermine the designated site's conservation objectives, in light of the characteristics and specific environmental conditions of the SAC/SPA/Ramsar. The likely significant effect test must be based on objective information and the risks must be real, not hypothetical.
- 5.3 Any measures that have been included in the application to reduce impacts on the European site cannot be considered within this test because they are included in the application as mitigation and would not be included were there not a need to mitigate this impact on European sites (see para 2.3).
- 5.4 Tables 1 and 2 on the following pages show the potential pathways through which impacts could arise through development of the project on the European nature conservation sites.
- 5.5 In order to determine the in-combination effects of the proposed development as

part of the wider developments around Hamble-Le-Rice, the Habitats Regulations Assessment for the Eastleigh Borough Local Plan 2016-2036 was consulted. The initial review focused on the in-combination assessment with other plans and projects to evaluate the scope of the assessment. This review confirmed that the in-combination assessment included the Joint Hampshire Minerals and Waste Plan (adopted 2013).

- 5.6 The Eastleigh Borough Local Plan 2016-2036 identifies 4.7ha of land at Mercury Marina and the Riverside Camping and Caravan Park off Satchell Lane for a marina, hotel, a range of holiday accommodation and car parking and boat storage (policy HA2). The policy notes that a site level Habitat Regulation Assessment is required to demonstrate how the site will be delivered without adverse effect on any European site.
- 5.7 Policy HA3 covers the restoration of the Hamble Airfield after mineral extraction and states that the site shall be restored in accordance with the Hampshire Minerals and Waste Plan and shall be retained as an area of accessible countryside with grazing, public access and outdoor recreation facilities laid out to the satisfaction of the Borough Council.
- 5.8 A search of current planning applications on the Eastleigh Borough Council website using the terms "Mercury Marina, Riverside Camping and Caravan Park and Satchell Lane", showed the last planning applications relating to the HA2 site were determined in 2019. An application by Foreman Homes to construction 61 dwellings immediately adjacent to the application site was refused in August 2021. Other recent applications largely relate to modifications to existing properties along Satchell Road or works associated with the operational marinas along the River Hamble.
- 5.9 There appears to be no projects in the immediate vicinity of the site that currently have the potential to act in-combination with this proposal. Should an application be made for redevelopment of Mercury Marina, in line with Policy HA2, during the operation of the quarry, the site level Habitat Regulation Assessment will need to demonstrate how the site can be delivered without adverse effect on any European site. At this stage without any firm details pertaining to the redevelopment of this HA2 site it is impossible assess any potential in-combination effects.
- 5.10 For the assessment of in-combination issues that, while likely to be insignificant at an individual project level, could have an adverse impact on the interest features of the NSN sites when considered in-combination with other plans or projects, this assessment relies on the conclusions of the Habitats Regulations Assessment for the Eastleigh Borough Local Plan 2016-2036.

## Table 1: Solent & Southampton Water SPA & corresponding Ramsar site and Solent and Dorset Coast SPA – assessment of likely significant effects

- $\sqrt{}$  Likely significant adverse effect on the Natura 2000 site
  - The principle is not relevant to the screening exercise

-

- x Not likely to have a significant adverse effect on the Natura 2000 site
- ? Uncertain effect on the Natura 2000 site

Check list of	Reduction	Direct effects	Indirect effects on the	Changes to the composition of the	Interruption or degradation of				
change	in area of	on the	populations of species for which	habitats for which the site was	the physical, chemical or				
	Annex 1	populations of	the site was designated or	designated (e.g. reduction in	biological processes that support				
	habitats?	species for	classified due to loss or	species structure, abundance or	habitats and species for which the				
Potential \		which the site	degradation of their habitat	diversity that comprises the	site was designated or classified?				
impacts		is designated	(quantity/quality)?	habitat over time)?					
Land take	-	Χ	X	-	-				
	The site lies o	utside the bounda	ry of both of the SPAs. No land take	within the SPAs is required, and no dir	ect impacts on populations for which				
	the SPAs are	classified will occ	cur.	•					
	The site doe	s not currently c	ontain any habitats suitable for fo	raging terns from the SDC SPA.					
		5	5	6 6					
	There is no e	evidence to indic	ate that the site supports population	ns of birds associated with the SSW	SPA/Ramsar It is not considered				
	that this site	is functionally l	inked to the SSW SP $\Delta$ /Ramsar T	here is no requirement in Policy $HA$	3 of the Eastleigh Borough Local				
	Plan to resto	re the site to hal	pitate that would be suitable for u	a by SDA spacies	is of the Eastergh Dorough Local				
	I fail to resit		stats that would be suitable for d	se by SIA species.					
	The Habitat	Deculations Ac	accompany for the Eastleigh Densus	h Local Dian concluded that there a	une no libely significant offect on				
	The Habitat	Regulations As	sessment for the Eastleigh Boroug	In Local Plan concluded that there v	was no likely significant effect on				
	land supporting habitat outside of the boundary of the SSW SPA/Ramsar from the plan alone or in-combination with other plans								
	and projects. The other plans and projects included in the Joint Hampshire Minerals and Waste Plan (adopted 2013).								
	Conclusion:	No likely signific	ant effect alone or in combination	with other plans and projects.					
				I					
	-	?	2	-	-				

Check list of	Reduction	Direct	effects	Indirect	effect	S OI	n the	Changes t	to the co	ompositio	n of the	Interruption or degradation of
change	in area of Anney 1	0n nonulat	the	population	s of sp	ecies fo designs	or which	habitats i designate	torwhi d (eg	ich the s	ite was	the physical, chemical or biological processes that support
	habitats?	species	for	classified	due	to l	oss or	species st	tructur	e. abunda	ance or	habitats and species for which the
Potential		which t	he site	degradatio	on of	their	habitat	diversity	that	compris	es the	site was designated or classified?
impacts \		is design	nated	(quantity/	quality)	?		habitat ov	ver time	e)?		
Wintering	The nature of	f the devel	opment	requires rela	tively fe	w full-t	ime staff	to be presen	t on site	e. It is estir	nated tha	t at the busiest periods approximately
birds:	7 people will	be worki	ng on the	e site. Drive	rs pickir	ng up m	inerals or	delivering i	nert res	toration m	laterial w	vill be on site only for relatively short
increased	periods. The	re is limite	d access	to the River	Hamble	e in this	area with	the slipway	accessi	ble from S	latchell L	ane (at Mercury Marshes), providing
recreational	the closest riv	verside ac	cess. Ac	cess to the S	SW SPA	A/Rams	ar is also	possible at I	Hamble	Common	and alon	g the Solent Way in Hamble-le-Rice.
disturbance	All these are	as are alre	ady acce	essible to the	public	and the	very sma	ll number o	f potent	tial visits t	o these a	reas by staff working at the quarry is
	not considere	ed to be a	significa	int addition	o currei	nt basen	ne activit	y levels.				
	The former L	Jambla A	irfiald si	to is privoto	land on	dianot	allocated	for any for	n of pul	blic roorog	ntional us	a it does not include any designated
	nublic footna	the or right	interest si	v However	it is cur	u is not rently u	anocated	101 ally 1011	nublic /	local resid	dents for	recreation and dog walking purposes
	The landowr	iers have	made re	peated effor	ts to try	and ke	en tresnas	sers off the	e site by	installing	y fencing	along the boundaries although this
	fencing has h	been conti	nually v	andalised by	the pul	olic. It i	s within t	he landown	er's righ	its to mak	e efforts	to prevent the public from accessing
	this land, suc	ch as insta	lling fen	cing barrier	s (inclu	ling mo	re robust	palisade tyr	be fenci	ng), and t	hev could	d choose to do this at any time in the
	future withou	ut requirin	g planni	ng permissio	on.			F		8),		
				01								
	Although the	use of thi	is site by	the public is	s technic	cally tre	spassing,	it is apparer	nt that c	urrently th	ie area is	regularly used for recreation and this
	will be curta	iled upon	the com	mencement	of the p	project.	However	, the propos	sals incl	ude the p	rovision	of a substantial permissive footpath,
	approximatel	ly 2.1 kilc	metres i	n length, ru	nning al	ongside	the north	nern and eas	stern bo	undaries o	of the site	e (within a generous stand-off buffer
	zone of retain	ned habita	t betwee	n 14 and 10	3 metres	width)	and conn	ecting with	a numbe	er of exist	ing off-si	ite pedestrian routes. This will enable
	public comm	uting (on	foot), re	creation and	dog wa	lking ac	tivities or	n site throug	ghout the	e duration	of the op	perational phases and site restoration.
	Then once th	e quarryir	ng projec	t is complet	ed and s	ite resto	ration pla	in implemer	ited, the	e permissiv	ve footpa	th will be retained and there will also
	be a 'commu	inity acces	ss meado	w' provided	l in the f	ar north	n-east of t	he site whic	the pu	ublic will	be able to	o freely use as a recreational space in
	perpetuity.											
	There is the	notantic1 (	bot the -	moioat will	anco th	o nuhlio	/ 10001 ==	aidanta ta a	ook ovt	other area	og for reg	praction that may include name of the
	SPA/Ramear	• This cor	ild lead i	to the distur	hance of	e public f feedin	g or roost	ing birds /	Any cha	nges in pr	as IOI IEC	recreational activity are likely to be
	localised in t	nature as r	residents	are likely t	o seek c	nit othe	e or roosi r local eit	es rather th	an trave	l long die	tances to	reach alternative recreational areas
	Hamble Corr	imon and	the Sole	nt Way foot	path in H	Hamble	Le-Rice	are consider	red to be	e likely alt	ernative	locations, however Ordnance Survey

Check list of	Reduction	Direct effects	Indirect effects on the	Changes to the composition of the	Interruption or degradation of					
change	in area of	on the	populations of species for which	habitats for which the site was	the physical, chemical or					
	Annex 1	populations of	the site was designated or	designated (e.g. reduction in	biological processes that support					
Detential	habitats?	species for	classified due to loss or	species structure, abundance or	habitats and species for which the					
impacts		is designated	(augradation of their habitat	habitat over time)?	site was designated of classified:					
Impacts	mans show that there is also a considerable local network of other public footpaths/rights of way and cycle paths to the immediate north and									
	maps snow that there is also a considerable local network of other public footpaths/rights of way and cycle paths to the immediate north and west of the site which run through areas of land that are a substantial distance from the coastline and it is very likely that these would also be									
	utilised as an	alternative.	agin areas of faire that are a substantic		very likely that these would also be					
	<b>Conclusion:</b>	Impacts related	to displaced recreational activity	require appropriate assessment of	lue to the inclusion of embedded					
	mitigation (	in the form of a	permissive footpath around the	perimeter of the site (operational	and post restoration phases) and					
	'community	access meadow'	(post restoration phase only)) as pa	rt of the scheme - see Section 6.						
					N/					
Breeding	-	-		X	X					
birds:	The breeding	g colonies within t	he SSW SPA are located a significant	t distance from the development for the	here to be to no likely impacts to the					
Increased	breeding bird	is. The HBIC data	a search did not reveal any records v	within 500 metres of the site for any	of the breeding citation features.					
disturbance	The alogast t	orn colony is at Tit	the state of the second approximately ap	vinetaly 0.2 kilomatars from the prop	and davelopment area. Tern apacies					
uistui bance	are opportun	istic feeders and t	heir diet consists predominantly of sn	all fish and occasionally planktonic of	rustaceans and insects. Though they					
	can forage fo	or up to 37 kilome	ters from their nesting sites, the propo	used development area contains no ha	hitat suitable for nesting or foraging					
	for tern speci	ies. (Cramp and Si	mmons, 1985, BirdLife International	2000).	stat suitable for nesting of foraging					
	Given the dis	stance from the de	velopment and the inaccessibility of b	reeding sites no likely significant effe	ects are predicted.					
	Access to the	e water's edge is h	highly unlikely to affect foraging tern	s. Both common and sandwich terns	will forage in shallow water close to					
	areas where there are high levels of human activity. This is apparent within the SSW SPA/Ramsar where common terns will fish pools alongside									
	the seawall a	t Pennington and s	sandwich terns foraging along the sho	reline of Studiand Bay and Pool Harb	our.					
	Conclusion: No likely significant effect alone or in combination with other plans and projects									
	Conclusion.	i to incery signific	and effect alone of in combination	the other plans and projects.						
	-	?	?	X	X					

Check list of	Reduction	Direct effects	Indirect	effects	on tl	e Changes	to the co	omposition of the	Interruption or degradation of
change	in area of Anney 1	on the	populations	of specie was desi	s for which onated	h habitats	for whi ed (e.g.	ch the site was	the physical, chemical or biological processes that support
	habitats?	species for	classified	due to	loss o	or species s	structure	abundance or	habitats and species for which the
Potential		which the site	degradation	n of the	eir habit	at diversity	that	comprises the	site was designated or classified?
impacts \		is designated	(quantity/q	uality)?		habitat o	over time	e)?	
Disturbance	The works w	ill be located appr	roximately 28	7 metres f	rom the ne	arest point of	the SSW	SPA. For the mo	ost part, the SSW SPA boundary is at
of wintering	least 300 me	tres from the appl	ication site be	oundary an	d is screen	ed from the S	SSW SPA	A by existing built	t development, hedgerows, scrub and
& breeding	woodland. T	here are no direct	views betwee	en the east	ern side of	the application	on site an	id the SSW SPA,	this was confirmed on the ground by
birds from	one of LCES	ornithologists du	iring a recent	fieldwork	v1s1t on 09	/11/2021.			
quarrying	Noise and vi	sual impacts have	been ruled o	ut nrimari	v due to tl	e distance of	f the proi	ect from the SPA	/Ramsar The University of Hull has
activities	produced a V	Waterbird Disturb	ance Mitigation	on Toolkit	to inform	estuarine pla	nning and	d construction pro	piects (Cutts et al 2013) The toolkit
	provides info	ormation on specie	s' responses	to varying	noise level	s and sources	of visual	l disturbance.	
	1	1	1						
	Overall, the t	toolkit concludes f	that noise lev	els below :	50dB prom	oted a low-le	evel respo	onse in most estua	rine species covered in the toolkit. A
	low level of r	response is classed	as one where	there is ur	likely to b	e an observab	le respon	se to the noise, e.g	g. reduction in feeding, birds scanning
	for danger et	c. It should be no	ted that an o	bservable 1	eaction in	a bird specie	s is not t	he same as an im	pact. A brief change in behaviour in
	response to a	. noise event will r	iot necessaring	y nave any	impact on	the individua	u(s) conc	erned.	
	The toolkit s	uggests that the m	ost sensitive s	species of y	wader will	demonstrate :	an alert re	esponse to certain	forms of visual disturbance at ranges
	of approxima	ately 300 metres.	In certain cire	cumstances	(in count	ies where br	ent geese	e are a quarry spe	cies) brent geese have been recorded
	responding to	o disturbance stim	uli at ranges o	of 350 met	æs.		U	1 2 1	,
			-						
	There are no	breeding tern cold	onies within 1	kilometre	of the site	boundary, so	noise and	d visual disturban	ce is not considered to be an issue for
	these species	. Foraging terns ra	inge over wid	e distances	and their	use of the Hai	mble will	vary temporally a	and spatially in response to a range of
	factors such a	as the state of the	tide, presence	of fish and	i season.				
	As the applic	ation area is mainl	v at least 300	metres fro	n the SSW	SPA and is s	screened f	from the SSW SP	A by existing residential development
	hedgerows. s	crub and woodlan	d. operational	auarrying	noise and	visual disturb	ance is th	erefore not antici	bated to have any impact on wintering
	or breeding S	SPA species.	, <b>r</b>	1 70					
	c	-							

Check list of change	Reduction in area of Annex 1	Direct effects on the populations of	Indirect effect populations of s the site was	cts on the pecies for which designated or	Changes to the habitats for w designated (e.	composition of the hich the site was .g. reduction in	Interruption or degradation of the physical, chemical or biological processes that support
Potential	habitats?	species for which the site	classified due degradation of	to loss or their habitat	species structu diversity that	re, abundance or comprises the	habitats and species for which the site was designated or classified?
impacts \		is designated	(quantity/quality	y)?	habitat over tin	ne)?	
	However, the	e application prop	oses that earth bunc	ls are constructed a	around the perimet	ter of the operational	quarrying site from the outset of the
	project in or	der to screen the v	vorks and provide	acoustics mitigation	on. There will also	be a generous stan	d-off buffer zone of between 14 and
	newly create	d habitats most n	otably retained and	newly planted he	differences along the	eastern boundary w	which will contribute to further visual
	screening he	re. Although not	included in the pro	posals specifically	to provide mitiga	ation for SPA birds,	the noise assessment does take into
	account the p	presence of this fea	ture when assessing	g noise levels on se	nsitive human rec	eptors outside the sit	e boundary. It is therefore considered
	to represent	mitigation and the	assessment of imp	acts relating to noi	se should be const	idered in an appropr	iate assessment.
		<b>NT • •</b> •	• • /				
	Conclusion:	Noise impacts re	quire appropriate	assessment due to	o the inclusion of	embedded mitigati	on (in the form of perimeter bunds)
	as part of th	le scheme - See Si					
Hydrological	-	-	X		X		X
changes,	Contaminatio	on of surface wate	<u>r</u>				
including:					_		
• water	The hydrolog	gical consultant h	as confirmed that t	here are currently	no surface water	features within the	footprint of the site and there are no
quanty	surface wate	r links from the s	tration canacity of	the ground The s	ilt pond identified	In discharge to grou	s will be used to collect silt from the
<ul> <li>nows</li> <li>abstraction</li> </ul>	extracted mi	nerals The silt wi	ll be retained on si	te and used to back	fill voids created	1 by the extraction p	rocess as restoration works progress
• nutrient	The water in	the freshwater po	and will be used for	r mineral washing	, the water will be	e sourced from rainv	ater and egress of ground water. As
levels	the River Ha	mble is influence	1 by tidal input from	n the Solent and fi	eshwater inputs fi	rom upstream, at thi	s point it is not dependant on ground
	water to main	ntain flows.					
	Conducion	No likely signifi	ant offerst element	n in combination	with other plana	an nucioata	
	Conclusion:	INU IIKEIY SIGNIII(	ant effect alone of	r in combination	with other plans	or projects.	
	Foul water						

Check list of	Reduction	Direct effects	Indirect	effects	on	the	Changes to t	the composit	ion of the	Interruption or degradation of
change	in area of	on the	population	s of spe	cies for	which	habitats for	which the	site was	the physical, chemical or
	Annex 1	populations of	the site	was d	lesignat	ed or	designated	(e.g. redu	ction in	biological processes that support
	habitats?	species for	classified	due	to los	ss or	species stru	cture, abun	dance or	habitats and species for which the
Potential		which the site	degradatio	on of	their	habitat	diversity t	hat compr	rises the	site was designated or classified?
impacts \		is designated	(quantity/o	quality)?			habitat over	time)?		
	This assessm	ent assumes that a	ll foul water	associate	ed with t	the site o	ffices and infr	astructure wi	ll be treate	d at Peel Common, a Southern Water
	facility and T	reatment Works.	The develop	ment cou	ld resul	t in incre	eased nitrogen	outputs to th	e SSW SP.	A/Ramsar and SDC SPA through the
	increase of fo	oul water that Peel	Common d	leals with	i, and th	e eventu	al discharge to	o sea via outf	falls into th	e Solent waters, which can cause an
	increase in nu	utrient loading (ni	rogen).							
	Current Natu	iral England guid	ance on nit	rogen ne		advises	local authorit	ties that com	mercial de	evelopment not providing overnight
	accommodati	ion should not gen	erally be req	uired to d	ieliver n	nitigatioi	n. This is to pro	event double	-counting	of waste water produced by residents
	inving and we	orking in the same	region.							
	Conclusion	No likely signific	ant effect a	lone or i	n combi	ination v	with other nls	ans and proj	ects	
	conclusion.	ito incery signific				mation	with other pit	ins and proj		
	<u>Flood risk:</u>									
	The site is loorivers or the site 3 to 5 metres of sand and g	cated in Flood Zo sea. The groundwa below ground surf gravels from the ce	ne 1, defined ater flood ris face across n lls.	d by the N sk at the s nost of the	National site is co e site on	Plannin onsiderec average	g Policy Fram I to be low giv ). The water ta	nework (NPP) yen the depth able may be e	F) as havin of ground xposed in p	g a low probability of flooding from water below the site (measured to be parts of the site through the extraction
	During both Hamble or So	operation and rest outhampton Water	oration phas through sur	ses there face wate	is no re er floodi	alistic in ing, fluv	npact pathway ial or coastal f	/ for any pos flooding or gr	sible pollut ound wate	ants from the site to enter the River r flooding.
	Following res and therefore attenuation p- to ground, all the infiltratio via surface w	storation, the perm e infiltration rates onds and conveya beit at the perimet n trenches is suffic vater or groundwat	eability of the across the nee structure er of the site ient to deal er is expected	he fill ma site are e es to perir e rather th with surfa ed.	terial us expected meter inf han with ace wate	sed to res to be 1 filtration nin it. Th r runoff	tore the site w ower. To miti- trenches. Thu he results from from the site.	till be lower the igate against is, the majorit in soakaway te Therefore, lit	han that of this a drai ty of rainfa ests have b tle net char	the sand and gravel reserve extracted nage scheme is proposed including Il to the site will continue to infiltrate een used to ensure that the design of oge in the balance of water discharged

change	Reduction	Direct effects	Indirect effect	ts on the	Changes to the composition of the habitats for which the site w	e Interruption or degradation of						
Change	Annex 1	populations of	the site was	designated or	designated (e.g. reduction	n biological processes that support						
	habitats?	species for	classified due	to loss or	species structure, abundance	r habitats and species for which the						
Potential		which the site	degradation of	their habitat	diversity that comprises t	e site was designated or classified?						
impacts \		is designated	(quantity/quality)	?	habitat over time)?							
	Conclusion: No likely significant effect alone or in combination with other plans and projects.											
	<u>Water abstra</u>	<u>ction</u>										
	During the operational phases, water collecting within the worked void will be pumped to other parts of the site where it will infiltrate to ground but the volumes of water requiring to be pumped will be small. A small proportion of the water that will collect within the lagoons constructed in the north of the site will be utilised for washing the mineral. Thus, little net change in the balance of water discharged via surface water or groundwater is expected during the operational phases of the site. Neither will additional abstractions to secure water supply to the site office affect the hydrology of the SPA habitats on which bird populations rely. Domestic water provision will be from Southern Water.											
	Conclusion	No Black offerst	lana an in aamhina	tion with other w	lang and musicate							
	Conclusion:	No likely effect a	lone or in combina	ntion with other <b>p</b>	plans and projects.							
Air quality	Conclusion:	No likely effect a	lone or in combina X	ntion with other p	plans and projects.	X						
Air quality changes	- The main po directly toxic	No likely effect a - llutants of concer effect upon veget	lone or in combina X n for NSN sites are cation.	tion with other p	<b>X</b> en (NOx), ammonia (NH <sub>3</sub> ) and su	X phur dioxides (SO <sub>2</sub> ). NOx can have a						
Air quality changes	- The main po directly toxic Sulphur diox of coal and o	No likely effect a - llutants of concert effect upon veget ide emissions are of il as well as (partic	lone or in combina X n for NSN sites are ation. overwhelmingly infl cularly on a local sc	e oxides of nitrog luenced by the ou ale) shipping. Ver	<b>Note:</b> <b>X</b> <b>Instant of power stations and industriations and and and and and and and and and and</b>	X phur dioxides (SO <sub>2</sub> ). NOx can have a l processes that require the combustion l by traffic.						
Air quality changes	- The main po directly toxic Sulphur diox of coal and o Ammonia em material incre	No likely effect a - llutants of concerned effect upon veget ide emissions are of il as well as (partion hissions are domin eases in SO <sub>2</sub> or NH	Ione or in combination X n for NSN sites are tation. overwhelmingly infl cularly on a local sc ated by agriculture, H <sub>3</sub> emissions will be	e oxides of nitrog luenced by the ou ale) shipping. Ver with some chemic	<b>X</b> en (NOx), ammonia (NH <sub>3</sub> ) and su         tput of power stations and industriation in the supplication of power stations and industriation is generated with the supplication of the suplication	X phur dioxides (SO <sub>2</sub> ). NOx can have a l processes that require the combustion l by traffic. ontributions. As such, it is unlikely that						
Air quality changes	Conclusion: - The main po directly toxic Sulphur diox of coal and o Ammonia em material increa In addition, g	No likely effect a - llutants of concert effect upon veget ide emissions are o il as well as (partion hissions are dominne eases in SO <sub>2</sub> or NH preater NOx or amo	Ione or in combina X n for NSN sites are ation. overwhelmingly infl cularly on a local sc ated by agriculture, H <sub>3</sub> emissions will be monia concentratior	tion with other p e oxides of nitrog luenced by the ou ale) shipping. Ver with some chemic associated with t	And projects. X en (NOx), ammonia (NH <sub>3</sub> ) and su tput of power stations and industria ty little sulphur dioxide is generate cal processes also making notable of his development. sphere will lead to greater rates of	X phur dioxides (SO <sub>2</sub> ). NOx can have a l processes that require the combustion l by traffic. ontributions. As such, it is unlikely that nitrogen deposition to soils.						

Check list of	Reduction	Direct effects	Indirect	effects	on	the	Changes to t	the com	position of	f the	Interruption or degradation of
change	in area of	on the	population	s of speci	es for w	hich	habitats for	r which	the site	was	the physical, chemical or
	Annex 1	populations of	the site	was des	signated	or	designated	(e.g. 1	reduction	in	biological processes that support
	habitats?	species for	classified	due to	loss	or	species stru	icture, a	abundance	e or	habitats and species for which the
Potential		which the site	degradatio	n of th	leir ha	bitat	diversity t	that co	omprises	the	site was designated or classified?
impacts	Tracffic	1s designated	(quantity/g	uality)?	4 <b>.</b>	. D22	habitat over	r time)?			
	The A27 are	lated with the pro	posal will ac	cess the si	te via the	B33	97 (Hamble La	Lane). In	ns road jon	ns the	A2/ at the windhover roundabout.
	The A27 cro	sses the River Har	note at Lowe	r Swanwic	Ж.						
	Natural Engl	and (2018) guida	nce documen	t Natural	Fnaland	's ann	worden to advis	isina con	nnotont aut	thorit	ies on the assessment of road traffic
	emissions un	der the Habitats F	Regulations e	xplains the	t it is wi	delv a	accepted that in	mpercen	tible impac	cts are	those which are less than 1% of the
	critical level	or load, which is	considered to	be rough	lv equiva	alent t	to 1.000 AAD <sup>-</sup>	T for car	rs and 200	AAD	T for HGVs. This was based on the
	Design Manu	ual for Roads and	Bridges (DM	RB) scree	ning tool	using	z Department f	for Trans	sport data t	to cale	culate whether the NOx output could
	result in a ch	nange of more that	n 1% of the	critical lev	el/load. '	The d	ocument also	suggests	s an initial	200 1	metre screening distance from roads
	when conside	ering impacts on N	<b>JSN</b> sites.								-
	Research pro	duced by AQC has	s highlighted	the need to	o also cor	nsider	the ammonia r	released	from vehic	cles w	'hen assessing the impact on nitrogen
	sensitive hat	ontats (Ammonia E	missions fro	m Roads	for Asses	ssing	Impacts on Ni	litrogen-	-sensitive F	Habite	<i>its</i> , AQC (2020). This is especially
	to nitrogen d	enosition and the	nositive effect	of reduc	ons nave	outpa of N	Ov in exhaust	is in ann	reducing ni	ssions	n deposition) is offset for ecological
	receptors by	the elevated level	s of ammonia	i or reduc	eu levels	5 01 IN	Ox III exilaust	l gases (I	leaueing in	uoge	il deposition) is offset for ecological
	receptors by		, or unmonit								
	At no point a	long the route bet	ween the Wi	ndhover R	oundabou	ut and	I the entrance t	to the sit	te is the SS	W SF	A less than 850 metres away. Along
	the A27 east	of the bridge cros	ssing the Har	nble at Bu	rsledon t	the SS	SW SPA is ove	er 300 m	netres from	the A	A27 at the closest point. Beyond the
	Hamble estua	ary traffic associat	ed with the pr	coposals w	ill be spre	ead ac	ross the wider	road net	twork as the	e dest	ination of lorries carrying aggregates
	will be deter	mined by regiona	l demand. Th	ne Habitat	Regulati	ions A	Assessment for	r the East	stleigh Bo	rough	Council Local Plan concluded that
	there were n	o likely significan	t effects from	n atmosph	eric poll	ution	from the plan	alone of	r in-combi	natio	n with other plans and projects. The
	other plans a	nd projects includ	ed in the Joir	it Hampsh	ire Minei	rals ai	nd Waste Plan	a (adopte	ed 2013).		
	SSW SPA is	largely unaffected	l hy nitrogen	denosite	APIS lists	s tern	s using coastal	l stable d	lune hahita	t as v	ulnerable to nitrogen deposition and
	common tern	using supralittora	l sediment as	vulnerabl	e to acid	denos	sition. Howeve	er, there a	are no bree	ding 1	terns in the vicinity of the application
	site and no si	uitable nesting hal	oitat (dunes o	r shingle h	eaches).			,			serve and the treating of the upproduction
			(								

Check list of	Reduction	Direct effects	Indirect ef	fects o	on the	Changes to	o the co	mposition of	the	Interruption or degradation of
change	in area of	on the	populations of	species f	or which	habitats for	or whi	ch the site v	as	the physical, chemical or
_	Annex 1	populations of	the site wa	s design	ated or	designated	l (e.g.	reduction	in	biological processes that support
	habitats?	species for	classified du	e to	loss or	species str	ructure	, abundance	or	habitats and species for which the
Potential		which the site	degradation	of their	habitat	diversity	that	comprises 1	he	site was designated or classified?
impacts		is designated	(quantity/qual	ity)?		habitat ov	er time	)?		-
_	<b>Conclusion:</b>	No likely signific	cant effect alone	and in co	ombination	n with other	<sup>.</sup> plans a	and projects		
	Dust generati	ion_								
	-									
	Dust and dirt	created by traffic	can be a problem	arising fro	om the ope	rations of cer	rtain typ	bes of develop	nent	, notably quarrying and the transport
	of quarried m	naterials. The guid	elines suggest th	at problem	ns with dus	st and dirt are	e unlike	ly to occur at o	lista	nces greater than 50 metres from the
	road. The imp	pact of dust and d	irt will depend o	n the mana	agement pr	actices under	rtaken o	on site.		
	At no point i	s the SSW SPA/F	Ramsar and SDC	SPA less	than 50 m	etres from th	ne site b	oundary. Due	to th	he distance between the site and the
	NSN sites no	potential impact	pathway is consi	dered to ex	xist.					
	Conclusion:	No likely signific	cant effect alone	and in co	ombination	n with other	r plans a	and projects		

#### Table 2: Solent Maritime SAC – assessment of likely significant effects

√ Likely significant adverse effect on the Natura 2000 site
 A The principle is not relevant to the screening exercise
 X Not likely to have a significant adverse effect on the Natura 2000 site
 Y Not likely to have a significant adverse effect on the Natura 2000 site

Check list of change Potential	Reduction in area of Annex 1 habitats?	Direct effects on the populations of species for which the site	Indirect effects on the populations of species for which the site was designated or classified due to loss or degradation of their habitat	Changes to the composition of the habitats for which the site was designated (e.g. reduction in species structure, abundance or diversity that comprises the	Interruption or degradation of the physical, chemical or biological processes that support habitats and species for which the site was designated or classified?
impacts \		is designated	(quantity/quality)?	habitat over time)?	
Land take	-	Χ	-	-	-
	The site lies of which the SA	outside the bound C is designated w	ary of the SAC. No land take within ill occur.	the SAC is required and no direct in	pacts on populations or habitats for
Increased	-	?	?	X	X
damage from	The River Ha	amble is approxim	ately 300 metres from the application	area at the closest point.	
recreational				_	
activity	As with the S	SPA and Ramsar, a	access to the SAC is most likely via N	Intercury Marshes off Satchell Lane. Ac	ccess to the SM SAC is also possible
	saltmarsh) ar	e relatively robus	t habitats that are unlikely be impact	ted by trampling In any case these	habitats occur within the dangerous
	intertidal zon	e and there is alw	ave a significant risk of people becom	ning stuck and/or sinking into the sof	t. wet sediments that comprise these
	habitats and t	then potentially dr	owning. It is considered that this cons	stant danger normally prevents most m	embers of the public from venturing
	onto saitmars	snes or mudifiats a	nd they will generally keep to designa	ued, wen-maintained lootpains.	
	Desmoulin's Desmoulin's surveys in 20	whorl snail is o whorl snail withi 009 and 2010. This	nly known to occur within Fishbou n the Solent Maritime SAC and the s s species is not present within the Har	rne Channel in Chichester Harbour. species was last recorded here in 2005 nble estuary (EBC HRA, 2018).	This is the only recorded site for 5. No individuals were found during

Check list of	Reduction	Direct effects	Indirect	effects	0n jes for v	the	Changes to the c	composition of the	Interruption or degradation of	
Change	Annex 1	populations of	the site	was de	esignated	or	designated (e.s	g. reduction in	biological processes that support	
	habitats?	species for	classified	due t	o loss	or	species structur	re, abundance or	habitats and species for which the	
Potential		which the site is designated	degradation	n of t wality)?	heir ha	bitat	diversity that habitat over tim	comprises the	site was designated or classified?	
	The application site will support a small number of full-time employees, it is estimated that at the busiest periods approximately 7 people will									
	be working on the site. Therefore, any visits to the SM SAC by these site personnel are likely to be very infrequent and no measurable direct									
	impact (trampling) deriving from the proposed development is forecast.									
	The former Hamble Airfield site is private land and is not allocated for any form of public recreational use, it does not include any designated									
	public footpaths or rights of way. However, it is currently used informally by the public / local residents for recreation and dog walking purposes.									
	The landown	ers have made rep	peated effort	s to try a	nd keep t	respas	ssers off the site b	y installing fencing	along the boundaries, although this	
	this land such as installing fencing barriers (including more robust palisade type fencing) and they could choose to do this at any time in the									
	future without requiring planning permission.									
	Although the use of this site by the public is technically transceing, it is apparent that autrently the area is recruled, used for represention and this									
	will be curtailed upon the commencement of the project. However, the proposals include the provision of a substantial permissive footpath,									
	approximately 2.1 kilometres in length, running alongside the northern and eastern boundaries of the site (within a generous stand-off buffer									
	zone of retained habitat between 14 and 103 metres width) and connecting with a number of existing off-site pedestrian routes. This will enable public commuting (on foot) recreation and dog walking activities on site throughout the duration of the operational phases and site restoration									
	Then once the quarrying project is completed and site restoration plan implemented, the permissive footpath will be retained and there will also									
	be a 'community access meadow' provided in the far north-east of the site which the public will be able to freely use as a recreational space in									
	perpetuity.									
	There is the potential that the project will cause the public / local residents to seek out other areas for recreation that may include parts of the									
	SAC and this	s could lead to trar	npling dama	ge of Ann	ex 1 hab	itats. A	Any changes in par	tterns of recreationa	l activity are likely to be localised in	
	and the Sole	nt Way footpath it	seek out othe n Hamble-Le	r local site -Rice are	consider	nan tra ed to	be likely alternativ	ve locations, howev	er Ordnance Survey maps show that	
	there is also	a considerable loc	al network c	of other pu	ublic foot	paths/	/rights of way and	cycle paths to the	ammediate north and west of the site	
	which run th	rough areas of lar	id that are a	substantia	al distanc	e fron	n the coastline and	l it is very likely th	at these would also be utilised as an	
	alternative.									

Check list of	Reduction	Direct effects	Indirect effects on the	Changes to the composition of the	Interruption or degradation of					
change	in area of	on the	populations of species for which	habitats for which the site was	the physical, chemical or					
	Annex 1	populations of	the site was designated or	designated (e.g. reduction in	biological processes that support					
Dotontial	nabitats?	species for	classified due to loss or	species structure, abundance or diversity that comprises the	nabitats and species for which the site was designated on classified?					
impacts		is designated	(quantity/quality)?	habitat over time)?	site was designated of classified:					
Impacts	is usignated (quantity/quanty). [nabitat over time).									
	Conclusion: Impacts related to displaced recreational activity require appropriate assessment due to the inclusion of embedded									
	mitigation (in the form of a permissive footpath around the perimeter of the site (operational and post restoration phases) and									
	'community access meadow' (post restoration phase only)) as part of the scheme – See Section 6.									
		1		1						
Disturbance	-	-	X	X	X					
from	The River Ha	amble forms the e	asternmost element of the Solent Ma	ritime SAC and is located approxima	tely 300 metres from the application					
construction	area at the closest point. The habitats within the SAC are not sensitive to noise and visual disturbance.									
activities	Demonstrate and the law of the law of the Field and Changelin Children and This is the law of the									
	Desmoulin's whort shall is only known to occur within Fishbourne Channel in Unichester Harbour. This is the only recorded site for Desmoulin's whort shall within the Selent Maritime SAC and the species was last recorded here in 2005. No individuals were found during									
	surveys in 2009 and 2010. This species is not present within the Hamble estuary (FBC HRA 2018)									
	surveys in 2007 and 2010. This species is not present within the Hamble estuary (LDC 1101, 2010).									
	Conclusion: No likely significant effect alone or in combination with other plans and projects.									
	Conclusion:	No likely signific	and effect alone of in combination	in other plans and projects.						
	Conclusion:	No likely signific								
Hydrological	-	No likely signific	X	X	X					
Hydrological changes,	- The same con	No likely signific	X for the SPAs and Ramsar sites – see d	X       liscussions above in Table 1.	X					
Hydrological changes, including:	- The same cor	No likely signific	X for the SPAs and Ramsar sites – see d	X liscussions above in Table 1.	X					
Hydrological changes, including: • water qual	- The same con	No likely signific - nclusion holds as f No likely signific	X         for the SPAs and Ramsar sites – see d         cant effect alone or in combination v	X         liscussions above in Table 1.         with other plans and projects.	X					
Hydrological changes, including: • water qual • flows	- The same cor Conclusion:	No likely signific - nclusion holds as f No likely signific	X for the SPAs and Ramsar sites – see d cant effect alone or in combination v	X liscussions above in Table 1. with other plans and projects.	X					
Hydrological changes, including: • water qual • flows • abstraction	- The same con Conclusion:	No likely signific	X for the SPAs and Ramsar sites – see d	X liscussions above in Table 1. with other plans and projects.	X					
Hydrological changes, including: • water qual • flows • abstraction • nutrient	- The same cor Conclusion:	No likely signific - nclusion holds as f No likely signific	X for the SPAs and Ramsar sites – see d cant effect alone or in combination v	X liscussions above in Table 1. with other plans and projects.	X					
Hydrological changes, including: • water qual • flows • abstraction • nutrient levels	- The same con Conclusion:	No likely signific	X         for the SPAs and Ramsar sites – see d         cant effect alone or in combination v	X liscussions above in Table 1. with other plans and projects.	X					
Hydrological changes, including: • water qual • flows • abstraction • nutrient levels Air quality changes	- The same con Conclusion: - Air quality	No likely signific - nclusion holds as f No likely signific -	X         for the SPAs and Ramsar sites – see d         cant effect alone or in combination v         X	X         liscussions above in Table 1.         with other plans and projects.         X	X					

Check list of	Reduction	Direct effects	Indirect	effects	on	the	Changes to the	he composit	tion of the	Interruption or degradation of
change	in area of	on the	population	s of speci	ies for w	hich	habitats for	which the	site was	the physical, chemical or
	Annex 1	populations of	the site	was de	signated	or	designated	(e.g. redu	iction in	biological processes that support
	habitats?	species for	classified	due t	o loss	or	species struc	cture, abun	idance or	habitats and species for which the
Potential		which the site	degradatio	n of ti	heir ha	bitat	diversity th	hat compr	rises the	site was designated or classified?
Impacts	Data francist	Is designated	(quantity/d				nabitat over	time):		A C is seen a line the large s on the f
	Data from the APIS website snows that currently the average critical load for nitrogen deposition across the SAC is exceeding the lower end of the range given for the two most constitue behitter. Derennial vagetation of story banks and shifting during slope the shoreline with Americanities and shifting during slope the slope the shoreline with Americanities and shifting during slope the slope t									
	une range given for the two most sensitive nabilats - Perennial vegetation of stony banks and snitting dunes along the shoreline with Ammophila aronario "white dunes". The range given for this babitat is between 8, 15kg/M/ba/yr and 10, 20kg/M/ba/yr respectively. Critical Leads are defined									
	archana while dunes. The range given for this haddat is between 6-15kg/lv/ha/yr and 10-20kg/lv/ha/yr respectively. Childal Loads are defined									
	as. a quantitative estimate of exposure to one of more ponutants below which significant narmin effects on specified sensitive elements of the environment do not occur according to present knowledge"									
	environment do not occur according to present knowledge.									
	Traffic assoc	ciated with the pro	posal will ac	cess the s	ite via the	e B33	97 (Hamble La	ane). This ro	ad joins the	e A27 at the Windhover roundabout.
	The A27 crosses the River Hamble at Lower Swanwick. At no point is the B3397 less than 850 metres from the SM SAC. The A27 crossing									
	the SM SAC at Lower Swanwick. At this point the site comprises the sub-tidal riverbed and an area of saltmarsh downstream of the $A27$									
	crossing at Hacketts Marsh. The inter-tidal mudflats in this area are outside the designated site.									
	Across the SAC the maximum level of nitrogen deposition is 18.5 kg/N/ha/yr with the average deposition being 11.6kg/N/ha/yr. For those									
	on the APIS	website for these	= A27 CIUSSII habitats is no	ig. estualle	zs allu At	d Im	sait meadows t	habitats are	only likely	to occur from increased traffic flows
	on the A27. The contribution of nitrogen deposition from traffic associated with the development is unlikely to result in any significant effects									
	on the interest features of the SAC closest to the A27 crossing.									
	As set out in Table 1 the B3397 is too far from the designated site for emissions from traffic associated with the proposals to be a likely impact									
	pathway.									
	Beyond the Hamble estuary, traffic associated with the proposals will be spread across the wider road network as the destination of lorries									
	carrying aggregates will be determined by regional demand.									
	The Habitat	Regulations Asses	sment for the	e Eastleigh	n Borougł	n Cou	ncil Local Plan	concluded t	that there w	vere no likely significant effects from
	atmospheric	pollution from the	e plan alone	or in-com	bination v	with o	ther plans and	projects. Th	ne other pla	uns and projects were included in the
	Joint Hamps	hire Minerals and	Waste Plan (	adopted 2	013).		_	-	-	-

Check list of	Reduction	Direct effects	Indirect effec	ts on the	Changes to the composition of the	Interruption or degradation of				
change	in area of	on the	populations of sp	becies for which	habitats for which the site was	the physical, chemical or				
	Annex 1	populations of	the site was	designated or	designated (e.g. reduction in	biological processes that support				
	habitats?	species for	classified due	to loss or	species structure, abundance or	habitats and species for which the				
<b>Potential</b>		which the site	degradation of	their habitat	diversity that comprises the	site was designated or classified?				
impacts		is designated	(quantity/quality)	)?	habitat over time)?					
	Conclusion: No likely significant effect alone and in combination with other plans and projects.									
	<u>Dust generation</u> The same conclusion holds as for the SPAs and Ramsar sites – see discussions above in Table 1. <b>Conclusion: No likely significant effect alone or in combination with other plans and projects.</b>									

- 5.8 The analysis in Tables 1 and 2 established two potential pathways that could result in likely significant effects on the interest features of the SSW SPA and Ramsar the SDC SPA and the SM SAC sites acting alone and in combination with other schemes. These relate to noise disturbance to SPA species present along the River Hamble during the operation of the site and displaced recreational activity potentially impacting on birds and habitats along the River Hamble (within both the SSW SPA/Ramsar and the SM SAC).
- 5.9 In the absence of being able to consider mitigation measures at this stage of assessment, and with regard only to the potential impact shown above, the proposals are considered likely to have a significant effect both alone and in combination with other developments on these sites
- 5.10 This conclusion generates a requirement for an appropriate assessment to address the impacts deriving from the scheme on the integrity of each of these sites (see Section 6.0).

## 6.0 Appropriate assessment

6.1 Section 5 identified the possible pathways likely to have a significant effect on the SSW SPA and Ramsar, the SDC SPA and the SM SAC either alone or in combination with other projects. In the absence of mitigation, the analysis identified the need for appropriate assessment of the impacts of noise disturbance and displaced recreational activity to establish whether the proposals will have an adverse effect on the integrity of the NSN sites, either alone or in combination with other projects.

#### Impacts of noise on birds (SSW SPA and Ramsar and SDC SPA)

- 6.2 The screening exercise identified that due to the distance between the site boundary and the SPAs, the potential for disturbance due to noise is limited. However, the proposals will involve the creation of earth bunds around the periphery of the site. These bunds will provide noise mitigation and visual screening. There will also be a generous stand-off buffer zone of retained and created habitats, between 14 and 103 metres width, allocated between the application boundary and the outside edge of the proposed earth bunding.
- 6.3 The creation of these bunds is embedded mitigation provided to mitigate impacts on sensitive receptors (although not included in the scheme in response to advice from the ecologist to mitigate impacts on ecological receptors). The calculations undertaken for the noise assessment have shown that due to the distances between the site and the SPA/Ramsar sites, the calculated site noise levels are no more than 4dB(A) above background noise levels at the nearest assessment location to the site.
- 6.4 In this instance the background levels at three receptor locations along Satchell Lane varied between 44 and 48dB, La90, with the average ambient noise levels being in the range 44 to 53 dB LAeq. Noise modelling calculated site noise levels at these receptors during works would be between 45 and 47 dB Leq. These levels would be achieved with the perimeter bunding in place.
- 6.5 These levels are below the 50dB low response threshold identified for estuarine species. The receptors used in the noise assessment are residential receptors and are also closer to the site boundary than the SPA/Ramsar. The noise assessment also assumes a reasonable worst case, the various machines assumed to operate at the closest practical position of the proposed simultaneous extraction/infilling areas for each receptor. It has also been assumed that the plant items work 100% of each hour apart from the tipping of inert material into the extraction void which is assumed to take place 20% of each hour. It should be noted that although this scenario is a possibility, the operation of the plant in this position is unlikely to happen in practice and would be only for a limited period of time if it did occur. Site noise levels at the SPA/Ramsar sites would therefore generally be expected to be below the levels presented in the noise assessment through most of the life of the site.
- 6.6 The predicted changes in noise levels at the SPA/Ramsar sites will be below the

threshold for disturbance to occur to estuarine bird species. The perimeter bunding is necessary to ensure that the site can be worked whilst keeping noise levels within the limits based on current government guidance. With this mitigation in place it is concluded that there is no likelihood of an adverse effect on the interest features of SSW SPA/Ramsar and SDC SPA, either alone or in-combination with other plans and projects.

#### Impacts of displaced recreational activity on birds (SSW SPA and Ramsar)

- 6.7 The screening exercise identified a possible impact on birds using the SSW SPA/Ramsar for foraging and roosting arising from increased recreational pressure on surrounding areas as a result of the existing informal access to the former Hamble Airfield site being curtailed from the commencement of the proposed project. However, given the considerable availability of alternative public rights of way, as well as cycle paths, (identified from OS mapping) in the immediate surrounding area which run through land that is a substantial distance from the local coastline and SSW SPA/Ramsar sites, it is therefore considered an unlikely scenario that the majority of the existing recreational activity on site will suddenly be displaced to the local coastline as a result of the project and that actually the public/local residents will also choose to utilise these other rights of way that are situated away from the coastline as another alternative. Taking this into account it is therefore assessed that the risk and magnitude of any adverse effects on birds using the SSW SPA/Ramsar will be low.
- 6.8 To address the issue of displacement of current recreational users of the site (albeit technically trespassers) during the operational and restoration periods the scheme design includes the provision of a substantial permissive footpath, approximately 2.1 kilometres in total length, running along the northern and eastern boundaries of the site and linking to existing off-site pedestrian routes. The pathway will also be situated within a generous buffer zone of retained and enhanced habitat, ranging from between 14 and 70 meters width along the length of the path, which will make it a pleasant and attractive route for the public to use. In the post operational phase and after the site restoration plan has been implemented, the permissive path will be retained in perpetuity and there will also be a 'community access meadow' provided in the far north-east of the site which the public will be able to freely use as an open area of recreational space.
- 6.9 The generous provision of accessible land for informal recreation throughout the operational lifespan of the project and then in the post restoration phase after the project is completed significantly reduces the risk of residents being displaced to the local coastline to undertake regular daily exercise and dog-walking. It should be noted that there are no direct links from the site to the SSW SPA/Ramsar via public rights of way, so therefore the redistribution of recreational activity to the eastern and northern fringes of the site will not increase the risk of locals accessing the River Hamble on foot as no direct links exist.
- 6.10 Taking into account the factors discussed in section 6.7 and the mitigation detailed in sections 6.8 and 6.9, it is therefore concluded that there is no likelihood of an adverse effect on the interest features of SSW SPA/Ramsar, either alone or incombination with other plans and projects.

#### Impacts of displaced recreational activity on habitats (SM SAC)

- 6.11 The screening exercise identified a possible impact arising from increased trampling of Annex I habitats within the SM SAC as patterns of recreational activity in the local area alter as a result of the existing informal access to the former Hamble Airfield site being curtailed from the commencement of the proposed project. However, given the considerable availability of alternative public rights of way, as well as cycle paths, (identified from OS mapping) in the immediate surrounding area which run through land that is a substantial distance from the local coastline and SSW SPA/Ramsar sites, it is therefore considered an unlikely scenario that the majority of the existing recreational activity on site will suddenly be displaced to the local coastline as a result of the project and that actually the public/local residents will also choose to utilise these other rights of way that are situated away from the coastline as another alternative. It is also identified that any visiting public to the local coastline and SAC areas are unlikely to venture onto Annex I habitat types occurring within the intertidal zones due to the risk to life of getting stuck in or sinking into soft, wet sediments and drowning, and instead will keep to safe, well-maintained footpaths. Taking these factors into account it is therefore assessed that the risk and magnitude of any adverse effects on Annex I habitats will be low.
- 6.12 As discussed in paragraph 6.8, to address the issue of displacement of current recreational users of the site (albeit technically trespassers) the scheme design includes the provision of a substantial permissive footpath running along the northern and eastern boundaries of the site (operational and post restoration phases) and a 'community access meadow' in the far north-east of the site (post restoration phase only).
- 6.13 The generous provision of accessible land for informal recreation throughout the operational lifespan of the project and then in the post restoration phase after the project is completed significantly reduces the risk of residents being displaced to the local coastline to undertake regular daily exercise and dog-walking.
- 6.14 Taking into account the factors discussed in section 6.11 and the proposed mitigation detailed in sections 6.12 and 6.13, it is concluded that there is no likelihood of an adverse effect on the interest features of SM SAC, either alone or in-combination with other plans and projects.

## 7.0 Conclusions

- 7.1 The development at the former Hamble airfield could potentially cause disturbance to birds feeding or roosting along the River Hamble, part of the SSW SPA and Ramsar and SDC SPA. Embedded mitigation in the form of 3 - 5-metre high earth bunds and a stand-off buffer zone of between 14 and 103 metres width around the site perimeter will reduce off-site noise levels associated with the operational quarrying works on site.
- 7.2 This mitigation will reduce the changes in noise levels at the SSW SPA and SDC SPA to levels where disturbance to waders, ducks, geese and terns is highly

unlikely to occur. There is no potential for any in-combination effects with other plans and projects.

- 7.3 The proposals could potentially displace existing local residents using the site for informal recreation. Displaced recreational activity could potentially impact on birds and habitats along the River Hamble (within both the SSW SPA/Ramsar and the SM SAC). Embedded mitigation in the form of a substantial permissive footpath running along the northern and eastern boundaries of the site (operational and post restoration phases) and a 'community access meadow' in the far northeast of the site (post restoration phase only), will provide considerable areas for informal recreation throughout the lifespan of the project and in the post restoration phase.
- 7.4 This mitigation will reduce the potential for existing recreational users of the site to be displaced to other locations within the SSW SPA/Ramsar and SM SAC. With no significant changes in patterns of recreational activity within the SSW SPA/Ramsar and the SM SAC predicted as a result of the implementation of the project and its associated mitigation, no impacts on birds and/or habitat within the NSN sites is therefore predicted. There is no potential for any in-combination effects with other plans and projects.
- 7.5 The screening of the project has identified no other realistic impact pathways that could impact on the interest features of the SSW SPA/Ramsar, the SM SAC and the SDC SPA.
- 7.6 On this basis, it is concluded that the proposals will not have an adverse effect on the integrity of the designated sites identified above, either alone or in combination with other plans and projects.
- 7.7 As Competent Authority, HCC must undertake its own independent appropriate assessment. It may adopt this document as the Council's own provided the contents have been subject to professional and independent scrutiny to confirm the findings of the assessment presented.

## 8.0 References

Air Quality Consultants. (2020) Ammonia Emissions from Roads for Assessing Impacts on Nitrogen-sensitive Habitats.

Boggis vs Natural England. Case number: C1/2009/0041/QBACF. Royal Courts of Justice. 20 October 2009.

Briels and others v Minister van Infrastructuur en Milieu. (C-521/12). Judgement of the Court (Second Chamber). European Court of Justice. Published 15 May 2014.

Cutts, N., Hemingway, K. and Spencer, J., 2013, Waterbird Disturbance Mitigation Toolkit Informing Estuarine Planning & Construction Projects [Version 3.2]. Institute of Estuarine & Coastal Studies (IECS) University of Hull.

Cramp, S. (1985) <u>Handbook of the birds of Europe, the Middle East and Africa. The birds</u> of the western Palearctic vol IV: terns to woodpeckers. Oxford: Oxford University Press

Eastleigh Borough Council (2018) *Eastleigh Borough Council Local Plan 2016-2036* 

Grace and Sweetman v An Board Pleanala (C-164/17). Judgement of the Court (Second Chamber). European Court of Justice. Published 25 July 2018.

Hampshire County Council (2013). *Hampshire Minerals and Waste Plan.* 

King D (2010) <u>Solent Waders and Brent Goose Strategy 2010. Hampshire and Isle of</u> <u>Wight Trust. Curdridge.</u>

Ministry of Housing, Communities and Local Government (2019) <u>National Planning</u> <u>Practice Guidance: Appropriate Assessment.</u>

Ministry of Housing, Communities and Local Government (2021) <u>National Planning</u> <u>Policy Framework.</u>

Natural England (2020) <u>Advice on achieving nutrient neutrality for new development in</u> <u>the Solent region. Version 5.</u>

Natural England (2018) <u>Natural England's approach to advising competent authorities</u> on the assessment of road traffic emissions under the Habitats Regulations.

<u>People over Wind and Peter Sweetman v Coillte Teoranta (C-323/17) Judgement of the</u> <u>Court. European Court of Justice. Published 12 April 2018.</u>

Urban Edge Environmental Consulting (2019) <u>Habitats Regulations Assessment for the</u> <u>Eastleigh Borough Local Plan 2016-2036. HRA report for the Submission Plan. June</u> <u>2019.</u>

Whitfield, D (2020) <u>Solent Waders and Brent Goose Strategy Hampshire and Isle of</u> <u>Wight Wildlife Trust. Curdridge.</u> www.apis.ac.uk

https://magic.defra.gov.uk

https://www.eastleigh.gov.uk/



## Appendix I: Application site location and boundaries



## Appendix II: Method of Working Plans (overview)



### Appendix III: Site location in relation to statutory sites

Please note: The boundaries for statutory sites have been provided as digital data from Natural England (NE); this digital data is indicative not definitive. Paper maps produced by NE at the time the sites were designated show the official site boundaries.